Author's response to reviews

Title: Factors associated with antenatal mental disorder in West Africa: A cross-sectional survey

Authors:

Bola Ola (wobola@yahoo.com)
Jim Crabb (jcrabb1@nhs.net)
Adetokunbo Tayo (tokunbunmi@yahoo.com)
Selena Gleadow Ware (sgleadowware@yahoo.co.uk)
Arup Dhar (arupdhar2000@hotmail.com)
Rajeev Rajeev Krishnadas (rajeev.krishnadas@gmail.com)

Version: 3 Date: 2 November 2011

Author's response to reviews:

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is "First name Initial Last name" (e.g. Joe F Bloggs) - This has been done

Abstract: This should be identical in the manuscript and the submission system. At present, it is not - This has been updated including the title of the study.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Abbreviations: Please format these in to a paragraph. This has been done.

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements. Please also remove all visible vertical lines. - This has been formatted accordingly.